Author’s response to reviews

Title: Trachoma in Yunnan province of Southwestern China: findings from Trachoma Rapid Assessment

Authors:
Min Wu (ynwumin@126.com)
Zhu Lin Hu (hzl77@263.net)
Dan He (hd@11.com)
Wen Rong Xu (wr@11.com)
Yan Li (yl@11.com)

Version: 3 Date: 27 Mar 2018

Author’s response to reviews:

Dear editors,

Thank you for giving us one more opportunity to revise the manuscript "Trachoma in Yunnan province of Southwestern China: findings from Trachoma Rapid Assessment" (BOPH-D-17-00107) . We read carefully the editor’s comments and made revision according to the suggestions. The point-by-point response to the comments is listed below. We hope the revision fulfills the requirement of BMC Ophthalmology and could be accepted. Thanks again!

Sincerely yours,

Min Wu

On behalf of all co-authors
1. Please provide all authors’ emails on title page.

Yes, we added all the authors’ emails on title page.

2. Abstract, Subtitle "Purpose" should be replaced as "Background", and check the sentence, "and and"

“Purpose” was replaced as “Background”. The duplicate “and” was deleted.

3. The raw data sheet includes the patients age and gender, which is not appropriate to disclose without patients consent to publish these data. In this case, we suggest that you could delete the raw data, and state in "Availability of data and materials" that "The datasets data sets of this study are available from the corresponding author upon reasonable request."

Thank you. The raw data was deleted and the statement was added as suggested.

4. "Funding: Supported by Sight First China Action Project" should be "This study is supported by ...."

The sentence was modified as suggested.

5. "Supporting Information" should be "Figure legends". Please only include the legends of Figure 1 here, and Figure 1A and 1B should be merged in a single composite figure. and uploaded as a single figure file. Please provide the high resolution version of Figure.

The name was modified as suggested. And one high resolution version of Figure 1 was provided.